Author's response to reviews

Title: A small molecular agent YL529 inhibits VEGF-D-induced lymphangiogenesis and metastasis in preclinical tumor models in addition to its known antitumor activities

Authors:

youzhi xu (youzhixu08@gmail.com)
wenjie lu (eva63136@163.com)
Peng Yang (1585627771@qq.com)
Wen Peng (251241462@qq.com)
Chun-ting Wang (553693616@qq.com)
Man-li Li (273107786@qq.com)
Yan Li (395067922@qq.com)
Guo-bo Li (645328757@qq.com)
Na-na Meng (1134079675@qq.com)
Hong-jun Lin (dazhisklb@foxmail.com)
Li-xin Kan (1443716876@qq.com)
Si-ying Wang (245428988@qq.com)
Sheng-yong Yang (ysy_008@163.com)
Luo-ting Yu (ylting_cd@163.com)
Ying-Lan Zhao (zhaoyinglan@scu.edu.cn)

Version: 10 Date: 7 May 2015

Author's response to reviews: see over
Dear editor and reviewers,

We are grateful for your letter with regard to the comments for our manuscript entitled “A small molecular agent YL529 inhibits VEGF-D-induced lymphangiogenesis and metastasis in preclinical tumor models in addition to its known antitumor activities (MS: 9316125914519646)”. Many thanks for you and the reviewers for the valuable comments.

We have thoroughly revised and addressed all the issues of this manuscript. The detailed explanations are included in the manuscript marked in blue or in the following up section of this letter point by point. We are looking forward to hearing from you at your earliest convenience time. Thanks again for your reconsideration.

Best regards

Sincerely Yours,

Yinglan Zhao, Ph.D
State Key Laboratory of Biotherapy and Cancer Center, West China Hospital, West China Medical School, Sichuan University, 17#, 3rd Section, Ren min South Road, Chengdu, 610041, China.
Phone: +86-28-85503628;
Fax: +86-28-85503628;
E-mail: zhaoyinglan@scu.edu.cn
Editor: (Comments to the Author)

Reviewer #1:

(Minor essential revision)

Q1: There were perhaps still a few misunderstandings from the authors, for example they do not seem to have understood the point I was making about reducing the number of decimal places on some of their data, but this can be fixed at the editing stage.

A1: Thanks the reviewer for the comment. According to the reviewer’s suggestion, we revised the rounding the numbers with two-decimal marked in blue in the revised manuscript.